
diff -r 174cc952ad36 -r 84ed8d6143ea handouts/ho07.tex
--- a/handouts/ho07.tex	Wed Mar 08 00:11:49 2017 +0000
+++ b/handouts/ho07.tex	Wed Apr 19 11:41:39 2017 +0100
@@ -533,6 +533,14 @@
 \url{https://adnauseam.io}
 \end{center}
 
+\noindent
+And a paper that predicts ad-blockers will in the end win over anti-ad-blocking:
+
+\begin{center}
+\url{http://randomwalker.info/publications/ad-blocking-framework-techniques.pdf}
+\end{center}
+
+
 \end{document}
 
 http://randomwalker.info/teaching/fall-2012-privacy-technologies/?




